Author's response to reviews

Title: Towards a more efficient diabetes control in primary care: six-monthly monitoring compared with three-monthly monitoring in type 2 diabetes - The EFFIMODI trial Design of a randomised controlled patient-preference equivalence trial in primary care

Authors:

Paulien R Wermeling (P.Wermeling@umcutrecht.nl)
Maureen van den Donk (M.vandenDonk@umcutrecht.nl)
Kees J Gorter (K.J.Gorter@umcutrecht.nl)
G. Ardine de Wit (G.A.deWit@umcutrecht.nl)
Yolanda van der Graaf (Y.vanderGraaf@umcutrecht.nl)
Guy EHM Rutten (G.E.H.M.Rutten@umcutrecht.nl)

Version: 2 Date: 26 April 2010

Author’s response to reviews:

(1) The title page with names, addresses and e-mail addresses is added to the revised manuscript
(2) The abstract and the trial registration is added to the revised manuscript
(3) The name of the ethics committee is added to the methods section of the revised manuscript
(4) Looking at the scope of both BMC Public Health and BMC Family Practice we agree on transferring our manuscript to BMC Family Practice.